Author's response to reviews

Title: Vibratory stimulation increase the electro-cutaneous sensory detection and pain thresholds in women but not in men

Authors:
Lisbeth Dahlin (Lisbeth.Dahlin@ki.se)
Irene Lund (Irene.Lund@ki.se)
Thomas Lundeberg (thomas.lundeberg@lidingo.mail.telia.com)
Carl Molander (carl.molander@ki.se)

Version: 4 Date: 17 May 2006

Author's response to reviews:

Formatting changes:
Figure and table placement text has been removed
The table (one) is included in the text and has been placed at the end, after the figure legends. One potential problem with the table is that is so wide that landscape format is needed. It is OK for us if you must change the format to portrait format by using for instance smaller text (?).
Table title is above the table, now in bold.
Figures have not been changed. It do not see how they can be cropped further.
No other changes have been introduced

Thank you